Author's response to reviews

Title: Kaplan anastomosis of the ulnar nerve: a case report

Authors:

- Georgios Paraskevas (g_paraskevas@yahoo.gr)
- Christos Ch Gekas (gekaschris@yahoo.gr)
- Alexandros Tzaveas (tzaveas@hotmail.com)
- Ioannis Spyridakis (jaspyr@otenet.gr)
- Alexandra Stoltidou (alexmed501@yahoo.com)
- Parmenion Ph Tsitsopoulos (par_tsits@yahoo.gr)

Version: 6 Date: 7 April 2008

Author's response to reviews:

The abstract has been structured into the following sections:

- Introduction
- Case presentation
- Conclusions